Author's response to reviews

Title: Affinity for risky behaviors following prenatal and early childhood exposure to tetrachloroethylene (PCE)-contaminated drinking water: a retrospective cohort study

Authors:

Ann Aschengrau (aaschen@bu.edu)
Janice M Weinberg (janicew@bu.edu)
Patricia A Janulewicz (paj@bu.edu)
Megan E Romano (romanom@uw.edu)
Lisa G Gallagher (lisa.g.gallagher@alum.bu.edu)
Michael R Winter (mwinter@bu.edu)
Brett R Martin (brmartin@bu.edu)
Veronica M Vieira (vmv@bu.edu)
Thomas F Webster (twebster@bu.edu)
Roberta F White (rwhite@bu.edu)
David M Ozonoff (dozonoff@bu.edu)

Version: 4 Date: 9 September 2011

Author's response to reviews:

The following changes were made to the manuscript in accordance with the September 2, 2011 e-mail from the Production Team:

1. The abstract in the submission system was revised to match the abstract in the manuscript.
2. A functional link was provided for reference #30.
3. Yellow highlighting was removed.
4. "et al." was removed from all references.
5. Tables in the additional files were labeled S1-S8 and tables in the main manuscript were labeled 1 and 2. Note that Table 2 is one long table, not several different tables. Its purpose is to summarize the information in the additional files. The format of this table is the same as that of Tables S3-S8 in the additional files.

I was not able to remove the blank page between Tables 1 and 2 because of the orientation change from portrait to landscape. Perhaps you will be able to do it.